Author’s response to reviews

Title: Correlation of skin rash and overall survival in patients with pancreatic cancer treated with gemcitabine and erlotinib – results from a non-interventional multi-center study

Authors:

Benedikt Westphalen (cwestpha@med.lmu.de)

Tobias Kukiolka (tkukiolka@web.de)

Benjamin Garlipp (benjamin.garlipp@med.ovgu.de)

Lars Hahn (l.hahn@onko-herne.de)

Martin Fuchs (martin.fuchs@klinikum-muenchen.de)

Peter Malfertheiner (peter.malfertheiner@med.ovgu.de)

Marcel Reiser (mreiser@pioh.de)

Fabian Kütting (fabian.kuetting@uk-koeln.de)

Volker Heinemann (volker.heinemann@med.uni-muenchen.de)

Andreas Beringer (andreas.beringer@roche.com)

Dirk Waldschmidt (dirk-thomas.waldschmidt@uk-koeln.de)

Version: 2 Date: 13 Feb 2020

Author’s response to reviews:

Editor comments:

1. WE NOTICE THAT AUTHOR LARS HAHN IS MISSING FROM THE AUTHORS’ CONTRIBUTIONS SECTION. THE INDIVIDUAL CONTRIBUTIONS OF ALL AUTHORS TO THE MANUSCRIPT SHOULD BE SPECIFIED IN THE AUTHORS’ CONTRIBUTIONS SECTION. GUIDANCE AND CRITERIA FOR AUTHORSHIP CAN BE FOUND HERE: …

We had inadvertently forgotten Lars Hahn (LH) in the authors' contributions section. We now clarify that LH “acquired data” and “wrote, reviewed and/or revised the manuscript” (Authors’ contribution, p. 17 para 3).

2. AT THIS STAGE, PLEASE UPLOAD YOUR MANUSCRIPT AS A SINGLE, FINAL, CLEAN VERSION THAT DOES NOT CONTAIN ANY TRACKED CHANGES, COMMENTS, HIGHLIGHTS, STRIKETHROUGHS OR TEXT IN DIFFERENT COLOURS. ALL RELEVANT TABLES/FIGURES/ADDITIONAL FILES SHOULD ALSO BE CLEAN VERSIONS. FIGURES
We have uploaded a final clean version of the manuscript, figures and additional files as requested.

Reviewer comments:
None.